Author’s response to reviews

Title: Effectiveness and Safety of Ferric Carboxymaltose Treatment in Children and Adolescents with Inflammatory Bowel Disease

Authors:

Martin W Laass (martin.laass@uniklinikum-dresden.de)
Simon Straub (simon.straub@uki.at)
Suki Chainey (suki.chainey@viforpharma.com)
Garth Virgin (garth.virgin@viforpharma.com)
Timothy Cushway (timothy.cushway@viforpharma.com)

Version: 2
Date: 6 April 2014

Author’s response to reviews:

Dear editors, here we submit the version of our manuscript with the requested changes and editorial requirements:

1. Line and page numbers are now included.

2. Regarding the ethic committee: The following sentence was added (page 8, line 5&6): "The study was approved by the local ethic committee of the Technische Universität Dresden (EK 411122010)." In the next sentence the word approval was replaced by permission: "Ethical permission for collection of anonymous data was granted prior to transcription from patient medical records to a case report form."

3. E-mail address of each author is now in bold letters in the title page.

I hope, our manuscript is now ready for the peer-review process.

Thank you verity much for your help.

Sincerely yours

Martin Laass
(on behalf of all authors)

Martin W. Laass, MD
Department of Pediatric Gastroenterology and Hepatology
Children’s Hospital, Technical University Dresden
Fetscherstr. 74, 01307 Dresden, Germany
Tel.: (+) 49-351-458-2267
Fax: (+) 49-351-458-5777
E-mail: martin.laass@uniklinikum-dresden.de